Author’s response to reviews

Title: Emotional exhaustion and burnout among medical professors; a nationwide survey

Authors:

- Joeri K. Tijdink (j.tijdink@vumc.nl)
- Anton C.M. Vergouwen (a.vergouwen@slaz.nl)
- Yvo M. Smulders (y.smulders@vumc.nl)

Version: 4 Date: 3 April 2014

Author’s response to reviews: see over
Cover Letter

Amsterdam, April 2014

Dear Editor,

On behalf of my co-authors, I would like to resubmit our manuscript “Emotional exhaustion and burnout among medical professors; a nationwide survey” as an original research article to BMC medical education after major revision.

The reviewers’ comments have been very valuable. We paid special attention to questions raised concerning the definition of burnout and focus more on the key component of burnout, emotional exhaustion.

In the attachment we respectfully respond point by point to all the reviewers comments in italic and refer in the comments to possible changes made in the manuscript.

Thank you very much for reviewing this submission. If any questions arise during the process, I would be happy to provide more information.

On behalf of all authors,

Yours faithfully,

J.K. Tijdink, MD, Psychiatrist
Dept. Of Internal Medicine, VU University Medical Centre
The Netherlands

VU University Medical Centre
Department of Internal Medicine
PO Box 7057
1007 MB Amsterdam
The Netherlands

Email: j.tijdink@vumc.nl
Phone: +31204444444